

EO-1 Weekly Report


	Earth Observing – 1

	May 23, 2012 – May 29, 2012

	DOY
	144
	–
	150

	Mission Day
	4262
	–
	4268


EO-1 SPACECRAFT

Instruments

· Scheduled 126 science Data Collection Events (DCEs) this week 

· Received all images
· Performed Instrument calibration

· ALI Internal Calibration I at 149/00:29z

EO-1 MISSION OPERATIONS CENTER

Real-Time

· Maintaining nominal spacecraft state of health

· Monitoring passes with the TrollSat ground station for receiver and commanding lock

· It appears that passes around 21:00z with TrollSat lose receiver and commanding lock a few minutes into the pass. These passes go from the South East and pass near the sensor mask provided by TrollSat. We are setting up tests with TrollSat to determine if this is the only area where we lose the locks.

· Preparing to upload X-band target table containing TrollSat
· Creating scripts to transfer primary ASIST T&C and update the central file hub that we are building.
· Adding functionality to prepare the target ASIST system for primary use automatically

· Speedwheel is ready to be active backup again
· Researching all files received on ASIST systems to see if they can all be sent to the central file hub and then transferred all Realtime systems

Mission Planning
· Creating 4-scan lunar calibration for use next week

· Designing and creating single, slow scan, Hyperion only Lunar image

· Researching updates on MP files to determine if they can be gathered on the central file hub and then transferred to both MP systems
Flight Dynamics
· Creating 4-scan lunar calibration for use next week

· Designing and creating single, slow scan, Hyperion only Lunar Image

· Working with ASPEN technicians to rebuild lunar calibration as 3-scan

· Researching updates to FD files to determine if they can be gathered on the central file hub and then transferred to both FD systems
System Administration

· Supported the IONet scanning team for the 2nd quarter All Highs scan by temporarily enabling the scanning computers to traverse the EO-1 mission firewall. No high vulnerabilities were found.

· Re-enabled DNS settings for the primary and backup FEDS computers. Removing these settings caused issues with system generated email notifications and antivirus updates. Ideally the FEDS software would resolve hostnames based only on the /etc/hosts file, but the use of the EO-1 mission firewall and Open IONet firewall minimizes its benefit.

· Began populating the “EO-1 IT Security Controls Equip List 2012” spreadsheet with data at the request of the IT security team.

· Completed applying the recent QuickTime update on the 5 CNE systems that have it installed.

· Created backups for the ASIST computers, flight dynamics computers, ASPEN computers, countdown clock, data processing computers, FEDS computers, firewall, trending system, and MOC CNE computer.

· Created weekly tape.

GROUND AND SPACE NETWORK

Station Downtimes 

· SGS has had an equipment malfunction and latest indication is that it will be back up August 2012
· HGS is down, unknown time to be back up

Operational Discrepancies

· None
Number of Real-Time Supports

[image: image1.emf]Date DOY S/X-Band S-Band Only Non-DSMC TDRSS Daily Totals

23-May-12 144 5 5 1 0 11

24-May-12 145 5 5 0 0 10

25-May-12 146 4 8 0 0 12

26-May-12 147 4 5 1 0 10

27-May-12 148 5 9 1 0 15

28-May-12 149 5 4 1 0 10

29-May-12 150 6 3 0 0 9

34 39 4 0 77 Weekly Totals


UPCOMING EVENTS

· X-band target table upload
· Redesign of Lunar Calibration to 3-scan

· Single, slow, Hyperion lunar image
· TR2 X-Band pass testing
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		Date		DOY		S/X-Band		S-Band Only		Non-DSMC		TDRSS		Daily Totals

		23-May-12		144		5		5		1		0		11

		24-May-12		145		5		5		0		0		10

		25-May-12		146		4		8		0		0		12

		26-May-12		147		4		5		1		0		10

		27-May-12		148		5		9		1		0		15

		28-May-12		149		5		4		1		0		10

		29-May-12		150		6		3		0		0		9

		Weekly Totals				34		39		4		0		77
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